Author's response to reviews

Title: Mother's compliance following recommendations to breastfeed or withhold breast milk during rotavirus vaccination in North India: a randomized clinical trial

Authors:

Temsunaro Rongsen-Chandola (naro@sas.org.in)
Brita A Winje (brita.winje@fhi.no)
Nidhi Goyal (goyalnidhi2004@yahoo.co.in)
Sudeep S Rathore (sudeep098@gmail.com)
Madhu Mahesh: (madhumahesh14@gmail.com)
Rajat Ranjan (drrajatranjan@yahoo.com)
Alok Arya (alokarya1980@yahoo.com)
Farhana A Rafiqi (farhanai@hotmail.com)
Nita Bhandari (nita.bhandari@sas.org.in)
Tor A Strand (tors@me.com)

Version: 4 Date: 11 June 2014

Author's response to reviews: see over
POINT BY POINT RESPONSE TO THE COMMENTS

"3.2 Please give the exact content of each of the two recommendations, including the rationale given to mothers for withholding breastfeeding or breastfeeding immediately after vaccine administration. This is important for readers to understand which factors secured compliance. The discussion argues that mothers will follow any recommendations as long as they think they are doing the right thing for their infant. How was this explained in the intervention process?

It is important for authors to add specific detail of each recommendation within the description of the intervention in the main manuscript, or as an appendix. This was the intervention and I think that it is important for readers to understand how it was introduced to mothers, as this would have a high influence on acceptability."

We have now added the exact content of the recommendations in the manuscript under the “Enrollment and Intervention Delivery” of the Methods section.

Editorial requests:
1. Please include the date of registration with your trial registration details at the end of your Abstract.
   The date has been included under Trial Registration.

2. Please include the reference number given with ethical approval, with your ethics statement in the Methods section.
   The reference numbers have been added under Ethical Clearance of the Methods section.

3. Please include a list of abbreviations used and their meanings, after your Conclusions.
   The list of abbreviations has been added.

4. Please move your figure legend below the Acknowledgements section.
   Needful has been done.